Author’s response to reviews

Title: Phenobarbital versus Morphine in the management of Neonatal Abstinence Syndrome, a randomized control trial

Authors:

Fatemeh Nayeri (fnayeri@sina.tums.ac.ir)
Mahdi Sheikh (mahdisheikh@gmail.com)
Majid Kalani (mahduni@icloud.com)
Pedram Niknafs (Pniknaffs@yahoo.com)
Mamak Shariat (mshariat@tums.ac.ir)
Hosein Dalili (hoseindalili@yahoo.com)
Ahmad Reza Dehpour (dehpour@medscape.com)

Version: 7  Date: 28 April 2015

Author’s response to reviews: see over
Dear Sir/Madam

Thank you most sincerely for your kind consideration of our paper entitled "Phenobarbital versus Morphine in the management of Neonatal Abstinence Syndrome, a randomized control trial" with the manuscript number of "MS: 9986270081386787". Here we are submitting the changes made to the manuscript as requested.

**These changes were made to the manuscript as recommended:**

1. Conclusion was added after the discussion section.
2. Authors’ contributions section was updated using the suggested format.
3. CONSORT checklist was completed and is uploaded with the revised manuscript.

Thank you again for your great comments

Yours Faithfully

Mahdi Sheikh, M.D, Ph.D

Hosein Dalili, M.D